Author’s response to reviews

Title: A Rare But Treatable Cause of Recurrent Chest Pain - Ictal Chest Pain

Authors:
Ching Soong Khoo (chingsoongkhoo@gmail.com)
Dongah Lee (peony1003@gmail.com)
Kang Min Park (smilepkm@inje.ac.kr)
Byung In Lee (BILEE@paik.ac.kr)
Sung Eun Kim (epidoc@inje.ac.kr)

Version: 2 Date: 28 Nov 2019

Author’s response to reviews:
The patient’s face and other personal details have been blurred as per suggestion. However, we do think that it’s important still to look at the patient’s face or head to look for head or eye deviation, which may be of lateralizing value.